# How to Achieve a 50% Reduction in Nutrient Loads from Agricultural Catchments under Different Climate Trajectories?

Maarten Wynants<sup>1</sup>, Johan Strömqvist<sup>2</sup>, Lukas Hallberg<sup>1</sup>, John Livsey<sup>1</sup>, Göran Lindström<sup>3</sup>, and Magdalena Bieroza<sup>1</sup>

<sup>1</sup>Swedish University of Agricultural Sciences <sup>2</sup>Swedish Meteorological and Hydrological Institute <sup>3</sup>Swedish Meteorological and Hydrological Institute (SMHI)

June 23, 2023

#### Abstract

Under persistent eutrophication of European water bodies and a changing climate, there is an increasing need to evaluate mitigation measures for reducing nutrient losses from agricultural catchments. In this study, we set up a daily discharge and water quality model in Hydrological Predictions of the Environment for two contrasting agricultural catchments in Sweden to forecast the impacts of future climate trajectories on nutrient loads. The model predicted a slight increase in inorganic nitrogen (IN) and total phosphorus (TP) loads under RCP2.6, likely due to precipitation-driven mobilisation. Under RCP4.5 and RCP8.5, the IN loads were forecasted to decrease from 16%-26% and 21%-50% respectively, most likely due to temperature-driven increases in denitrification and evapotranspiration. No distinct trends in TP loads were observed. A 50% decrease in nutrient loads, as targeted by the European Green Deal, was backcasted using a combination of mitigation scenarios, including i) a 20% reduction in mineral fertiliser, ii) introducing cover crops, and iii) stream mitigation by increasing the size of floodplains and wetlands. Target TP load reductions could only be achieved by stream mitigation, which is likely due to legacy effects and secondary mobilisation within agricultural streams. Target IN load reductions were backcasted with a combination of stream mitigation, fertiliser reduction, and cover crops, wherein the required measures depended on the climate. Overall, the diverging responses of nutrients to climate change and mitigation scenarios indicate that water quality management needs to be tailored to the catchment characteristics, and to the spatial and time specific effects of climate change.

#### Hosted file

963945\_0\_supp\_11006666\_rvztrn.docx available at https://authorea.com/users/631151/articles/ 650639-how-to-achieve-a-50-reduction-in-nutrient-loads-from-agricultural-catchmentsunder-different-climate-trajectories

# How to Achieve a 50% Reduction in Nutrient Loads from Agricultural Catchments under Different Climate Trajectories?

- 3 M. Wynants<sup>1</sup>, J. Strömqvist<sup>2</sup>, L. Hallberg<sup>1</sup>, J. Livsey<sup>1</sup>, G. Lindström<sup>2</sup>, and M. Z. Bieroza<sup>1</sup>
- <sup>1</sup>Department of Soil and Environment, Swedish University of Agricultural Sciences, SE-75007,
   <sup>5</sup> Uppsala, Sweden.
- <sup>2</sup>Hydrological research department, Swedish Meteorological and Hydrological Institute, SE 60176, Norrköping, Sweden.
- 8 Corresponding author: Maarten Wynants (<u>maarten.wynants@slu.se</u>)

# 9 Key Points:

- Inorganic Nitrogen loads were forecasted to decrease in the future under Representative
   Concentration Pathways 4.5 and 8.5.
- Target reductions in Total Phosphorus loads could only be achieved by stream mitigation.
- Inorganic Nitrogen loads were backcasted to cumulatively decrease with fertiliser
   reduction, stream mitigation, and cover crops.

## 15 Abstract

Under persistent eutrophication of European water bodies and a changing climate, there is an 16 increasing need to evaluate mitigation measures for reducing nutrient losses from agricultural 17 catchments. In this study, we set up a daily discharge and water quality model in Hydrological 18 Predictions of the Environment for two contrasting agricultural catchments in Sweden to forecast 19 20 the impacts of future climate trajectories on nutrient loads. The model predicted a slight increase in inorganic nitrogen (IN) and total phosphorus (TP) loads under RCP2.6, likely due to 21 precipitation-driven mobilisation. Under RCP4.5 and RCP8.5, the IN loads were forecasted to 22 decrease from 16%-26% and 21%-50% respectively, most likely due to temperature-driven 23 increases in denitrification and evapotranspiration. No distinct trends in TP loads were observed. 24 A 50% decrease in nutrient loads, as targeted by the European Green Deal, was backcasted using 25 a combination of mitigation scenarios, including i) a 20% reduction in mineral fertiliser, ii) 26 introducing cover crops, and iii) stream mitigation by increasing the size of floodplains and 27 wetlands. Target TP load reductions could only be achieved by stream mitigation, which is likely 28 due to legacy effects and secondary mobilisation within agricultural streams. Target IN load 29 reductions were backcasted with a combination of stream mitigation, fertiliser reduction, and cover 30 crops, wherein the required measures depended on the climate. Overall, the diverging responses 31 of nutrients to climate change and mitigation scenarios indicate that water quality management 32 33 needs to be tailored to the catchment characteristics, and to the spatial and time specific effects of climate change. 34

## 35 Plain Language Summary

The European Union has set a target to reduce nutrient losses from agricultural areas by 50% in 36 2030 to improve the quality of its water bodies. However, we argue that climate change will have 37 a strong impact on nutrient dynamics, implying that the required actions for improving water 38 quality need to adapt over time. In this study, we simulated the future losses of two major nutrients, 39 inorganic nitrogen and total phosphorus, for two different Swedish agricultural streams. We also 40 modelled potential actions to reach the targeted 50% reduction in nutrient losses. The model 41 predicted that inorganic nitrogen loads will decrease under medium and bad climate change 42 43 pathways, but increase under the best climate change pathway. We found that targeted reductions in total phosphorus loads could only be achieved by mitigating streams through increasing the 44 space for floodplains and wetlands, making it a critical action. Targeted reductions in inorganic 45 nitrogen loads could be achieved by combining stream mitigation with a 20% reduction in mineral 46 fertiliser, and by protecting the soil with cover crops in winter. This study has shown how we can 47 use water quality models for identifying the required actions for reducing future nutrient loads in 48 49 agricultural streams.

## 50 1 Introduction

In the agricultural regions of Northern and Western Europe, over 80% of water bodies fail 51 to reach good ecological or chemical status, with many being in poor or bad condition according 52 to the Water Framework Directive (Kristensen et al., 2018). Despite nutrient input reductions since 53 the 1990s (Lu & Tian, 2017), the negative impacts from diffuse pollution on aquatic ecosystems 54 are still evident, frequently causing eutrophication and hypoxia in inland and coastal and lake 55 waters (Andersen et al., 2017). Decades of over fertilisation have led to the buildup of legacy 56 nutrients in agricultural unsaturated soils, groundwater, and river sediments (Basu et al., 2022; 57 58 Bouwman et al., 2013). Extensive drainage networks have been installed throughout European farming areas in the form of open ditches, straightened streams, and subsurface tile drains (Schultz et al., 2007). While these have improved crop growing conditions, the increased hydrological connectivity has also resulted into faster nutrient and sediment exports from agricultural catchments (Blann et al., 2009; Castellano et al., 2019). The combination of high nutrient inputs, legacy stores, and artificial drainage continue to negatively impact water quality and aquatic biodiversity (Andersen et al., 2017; Ulén et al., 2007).

The European Green Deal targets to reduce nutrient losses from agricultural areas by 50% 65 in 2030, which is envisioned by decreasing fertilisation by 20% (European Commission, 2020). 66 However, achieving these targets will also require a structured implementation of the locally most 67 effictive catchment mitigation measures (M. Bieroza et al., 2021). EU member states provide land 68 owners with financial incentives for mitigating diffuse nutrient pollution (Boeuf & Fritsch, 2016; 69 Wiering et al., 2020). These measures target three impact pathways: (1) reducing nutrient inputs 70 to fields, (2) reducing erosion and mobilisation of nutrients, and (3) intercepting mobilised nutrient 71 and sediment flows. Despite the large financial investments and efforts by land owners, 72 improvements in water quality are often not observed (Destouni et al., 2017; Wiering et al., 2020). 73 This can be partly explained by climatic variation and extreme weather (Mellander et al., 2018), 74 and by the widespread prevalence of nutrient legacies in agricultural catchments (Basu et al., 2022; 75 Frei et al., 2021) that override catchment mitigation efforts. Moreover, mitigation measures are 76 often implemented with inadequate sizes, locations, or designs, based on personal preferences of 77 landowners and financial drivers (Roley et al., 2016; Uusi-Kämppä et al., 2000), which does not 78 reflect the spatial variability in catchment processes that govern nutrient transport and removal 79 (Basu et al., 2023; Hallberg et al., 2022; Walton et al., 2020). There is thus a need for a catchment-80 specific evaluation of mitigation measures that are required to achieve set water quality goals and 81 integrate those in a decision support strategies before committing and investing in specific 82 measures (Hogan et al., 2023). However, many uncertainties remain about the spatial and 83 temporally varied impacts of climate change on nutrient loads (Bol et al., 2018; Zia et al., 2022). 84 In this context, the impacts of climate change on future nutrient loads can be forecasted using 85 process-based water quality models (Bartosova et al., 2019; Ockenden et al., 2017; Zia et al., 86 2022). Moreover, process models also allow to decouple catchment mitigation outcomes from 87 climatic variability (Grimvall et al., 2014) and thereby provide a pathway for backcasting 88 scenarios to reach the desirable future reduction in nutrient loads (Capell et al., 2021; Hankin et 89 al., 2019). 90

91 The overall objectives of this study were (i) to forecast the impacts of climate change on future nutrient loads and (ii) to backcast the targeted 50% reduction in nutrient loads with 92 mitigation measures across all impact pathways. Using HYPE (Hydrological Predictions of the 93 Environment) and an ensemble of downscaled future climatic predictions, we modelled the 94 impacts of climate change and catchment mitigation on nutrient loads in two Swedish agricultural 95 catchments with different climate, soil textures and land use. This study focussed on three 96 representative concentration pathways (RCPs): 2.6, 4.5, and 8.5, in the near future (2022-2035), 97 mid future (2050-2065) and distant future (2085-2100). The three mitigation measures we 98 modelled were i) a 20% reduction in mineral fertiliser, ii) introducing cover crops between growing 99 seasons, and iii) stream mitigation by increasing the size of floodplains and wetlands 100

## 101 2 Materials and Methods

102 2.1 Study catchments

103 Hestadbäcken catchment is located in central east Sweden, and Tullstorpsån catchment is located in south Sweden. Both study catchments are agriculturally-dominated, but differ in size, 104 cropping regimes, rainfall, and soil type (Table 1, Figure 1, Figure S1). Tullstorpsån is larger 105 overall, but also has a higher percentage of cropland with significant amounts of root crops (8.3%) 106 and spring crops (7.6%) besides its dominant autumn crops (57.1%). The soils are mostly loamy, 107 with smaller pockets of moraine. Hestadbäcken is smaller and is dominated by autumn crops 108 (54.7%) cultivated on clay soils. It also has larger areas of forest (25.5%) and pasture (16.9%), 109 which are mostly developed on the moraine soils. Tullstorpsån is on average wetter and warmer 110 111 compared to Hestadbäcken, which has less rainfall and a larger temperature range.

112

Catchment	Hestadbäcken	Tullstorpsån						
Catchment area (km2)	7.6	62.1						
Elevation range (m)	44-85	3-101						
Dominant land use types	Autumn crops (54.7%),	Autumn crop (57.1%), Pasture						
	Forest (25.5%), Pasture	(15.2%), Root crops (8.3%), Forest						
	(16.9%) (7.6%), Spring crops (7.1%)							
Floodplain and wetland area	$2500 \text{ m}^2 (0.03\%) \qquad 507,311 \text{ m}^2 (0.82\%)$							
Dominant soil classes	Moraine (29.2%), Silty clay	Loam (41.8%), Sandy loam						
	(23.7%), Clay loam (19.7%),	(29.6%), Clay loam (6.4%),						
	Clay (8.7%)	Moraine (6.0%)						
$\mu \pm \sigma$ yearly rainfall (mm)	$580 \pm 131$	$790 \pm 115$						
$\mu \pm \sigma$ of >15mm rainfall days year <sup>-1</sup>	$6.5 \pm 3.6$	$7.5 \pm 3.3$						
$\mu \pm \sigma$ of >30mm rainfall days year <sup>-1</sup>	$0.9 \pm 0.8$	$1.0 \pm 1.3$						
Temperature (µ and range in °C)	7.9; -12.8 to 23.3	8.7; -7.9 to 23.1						

113 **Table 1**: Overview of current catchment characteristics with  $\mu$  as mean and  $\sigma$  standard deviation.

A watershed assessment was performed for both study catchments with the hydrology 114 toolset in ArcMap 10.8.1 (ESRI, 2020) and based on the 2m-resolution Grid2+ digital terrain 115 model (Lantmäteriet, 2019). Land cover was determined by performing a maximum likelihood 116 classification on a 0.25m-resolution RGBI orthophoto from 2021 (Lantmäteriet, 2021) following 117 the methodology of Wynants et al. (2018) to the following land use classes: forest, pasture, 118 cropland, build-up areas, open water, stream, floodplains & wetlands, and rock outcrops. For each 119 field, the cropland class was subdivided in autumn crops, spring crops and root crops based on 120 2021 crop rotation information (Jordbruksverket, 2022). Soil texture classes were derived from the 121 50m-resolution Digital Soil Map of Sweden for arable land (Piikki & Söderström, 2019), the 122 Quaternary Deposits map from the Geological Survey of Sweden (Sveriges Geologiska 123 Undersökning, 2014), and from the land cover classification for build-up areas, rock outcrops, 124 water and stream classes. Rainfall and temperature data were obtained from the climate stations of 125 the Swedish Meteorological and Hydrological Institute (2023b) and from the PTHBV grid (Berg 126 et al., 2016; Swedish Meteorological and Hydrological Institute, 2023a). 127

128 2.2 Measurements of discharge and nutrient concentrations

Hestadbäcken was gauged continuously for discharge on sub-hourly time steps using a small basin with V-notch. The stream was also sampled flow-proportionally on a fortnightly interval, and samples were analysed for total phosphorus (TP), soluble reactive phosphorus (SRP), particulate phosphorus (PP), inorganic nitrogen (IN), total nitrogen (TN), and suspended sediment (SS) (Kyllmar et al., 2014). Flow-proportional nutrient concentrations were converted to daily
 estimates of nutrient loads using the average daily discharges.

Tullstorpsån was monitored intermittently at three different gauge locations (upstream, midstream, and downstream; Figure 1: Location of the study catchments, their land cover distributions and stream gauges.) for either water stage or water quality. Water stage sensors were placed upstream and midstream in May 2020. Continuously recorded water depths (S) were converted to stream discharge (Q) using stage-discharge relationships. In the upstream, we used an exponential relationship (Equation (1);  $R^2=0.831$ ), while in the midstream we used a split relationship, where an exponential function was followed by a linear (Equation (2)).

$$Q_{up} = 0.6863 \, x \, S_{up}^{2.813} \tag{1}$$

$$Q_{mid} = f(S_{mid}) = \begin{cases} 30.211 \ x \ S_{mid}^{4.3839} & S_{mid} < 0.384 \\ 3.208 \ x \ S_{mid} - 0.8418 & S_{mid} \ge 0.384 \end{cases}$$
(2)



142

143 **Figure 1**: Location of the study catchments, their land cover distributions and stream gauges.

144 2.3 Model set-up, calibration and validation

The catchment models were set-up in HYPE, which is a semi-distributed and open source hydrological and nutrient transport modelling framework. Detailed information on the set-up procedures, model assumptions, and model characteristics can be found in Text S1. For more information on the representation, parameterisation, and sensitivity of hydrological and nutrient

149 processes, we refer to Lindström et al. (2010), and Santos et al. (2022).

The catchment models were built on combinations of Soil type and Land use classes 150 (SLCs), with 27 SLCs in Hestadbäcken, and 60 SLCs in Tullstorpsån. Tullstorpsån was divided in 151 three sub-catchments based on locations of stream gauges. The models were forced with daily 152 average precipitation and temperature data. For each SLC, the soil system was classified with up 153 to three soil layers with defined depths. Specific crop types and tile drain depths were assigned to 154 the agricultural SLCs. Each crop type was assigned planting, harvesting, and ploughing dates, as 155 well as the amounts and dates of application of mineral fertiliser and manure, based on agricultural 156 monitoring programs at nearby catchments (Kyllmar et al., 2014). Besides fertilisation, nutrients 157 enter the catchment through atmospheric deposition and rural sewage, and leave through crop 158 uptake, denitrification, and with stream water discharges. Starting pools of nutrients in the soil 159 water, organic form, and bound to soil particles were specified for different SLCs and represent 160 legacy stores of nutrients (Strömqvist et al., 2012). Floodplains & wetlands were grouped in one 161 SLC with a defined fraction of the catchment runoff, wherein nutrient removal was modelled using 162 the 'wetland' subroutine with parameters controlling nutrient retention, denitrification, and 163 macrophyte uptake (Arheimer & Wittgren, 2002; Tonderski et al., 2005). Streams were modelled 164 as dynamic pools of sediment and PP, wherein sedimentation, erosion, and resuspension can 165 (im)mobilise sediment and PP (Bartosova et al., 2021). 166

The simulation results were evaluated by comparison with observations of daily discharge 167 and nutrient loads using a set of 'goodness of fit' measurements (Moriasi et al., 2015), i.e. Nash-168 Sutcliffe efficiency (NSE; Nash and Sutcliffe (1970)), Kling-Gupta efficiency (KGE; Gupta et al. 169 (2009)), Pearsons's correlation coefficient (R), and relative bias (RE; percentage of difference). 170 The model was optimised using the calibration methodology of Hundecha et al. (2020), wherein 171 we combined manual and automatic calibration based on the Differential Evolution Markov Chain 172 (DE-MC) algorithm of Braak (2006). The calibration was carried out following a stepwise 173 approach by identifying key parameter groups and calibrating these together within possible 174 ranges, while keeping other parameters fixed to reduce potential equifinality (Strömqvist et al., 175 2012). Because nutrient transport is largely governed by hydrology, the model was initially 176 calibrated for discharge, and subsequently calibrated for the water quality parameters. A split 177 sample was used to validate the model setups for uncalibrated periods. The model fit was evaluated 178 using the simplified model evaluation thresholds of Moriasi et al. (2015) and the evaluation 179 guidance for KGE as in Knoben et al. (2019). A simple sensitivity analysis was performed on 180 wetland parameters by measuring the NSE variation during Monte Carlo simulations with random 181 perturbations of parameters between set intervals (Santos et al., 2022). 182

## 183 **2.4 Future climate trajectories**

We used an ensemble of three general circulation models (GCM) from the Coupled Model 184 Intercomparison Project Phase 5 (CMIP5): the "Met Office Hadley Centre ESM, HadGEM2-ES" 185 model (Jones et al., 2011), the "Max Planck Institute ESM-LR" (Popke et al., 2013), and the 186 187 "ICHEC-EC-EARTH" (Hazeleger et al., 2010). These GCMs simulate representative concentration pathways (RCPs) 2.6 (stringent reduction), 4.5 (medium), and 8.5 (business as usual) 188 (Collins et al., 2013). As described in Jacob et al. (2014), the GCMs have been downscaled to a 5 189 km grid over the northern European regions using the "KNMI regional atmospheric climate model 190 (RACMO) version 2" (van Meijgaard et al., 2008) and the "SMHI Rossby Centre regional climate 191 model" (SMHI-RCA4) (Strandberg et al., 2015). An overview of the downscaled climate models 192 193 can be found in Table S1. Outcomes from these downscaled climate models were statistically scaled against a reference temperature and precipitation data-set using the distribution-based 194 scaling algorithms of W. Yang et al. (2010). The resulting daily temperature and precipitation data 195

were used in this study to model changes in nutrient loads under future climatic conditions. Data 196 197 analysis was performed in R (R Core Team, 2022). Annual yearly average rainfall, temperature, and amount of high rainfall days (>15mm per day & >30mm per day) for the period 2000-2022, 198 199 2022-2035, 2050-2065, and 2085-2100 were estimated using the aggregate function. We subsequently performed *t-tests* in R to 1) compare the model predictions with empirical 200 measurements for 2000-2022, 2) evaluate the difference between the climate models and 3) test if 201 the projected changes in rainfall and temperature were significant. Model calculations of daily 202 nutrient loads were summed to total yearly nutrient loads. For each period and RCP, basic statistics 203 (mean, median, standard deviation, and interguartile range) were calculated for both individual 204 models and the ensemble of models. T-tests were performed between the nutrient load predictions 205 of different models, RCPs, and periods to evaluate the differences. The percentage of change 206 between the periods was calculated to evaluate trends in nutrient loads. 207

#### 208 **2.5 Catchment mitigation scenarios**

Scenario	Abbreviation	Impact pathway	Description
Original	Original	/	In <u>Hestadbäcken</u> , ca. 2500 m <sup>2</sup> of floodplain & wetland
			(0.03% of catchment). In <u>Tullstorpsån</u> , ca. 0.51 km <sup>2</sup> of
			floodplain & wetland (0.82% of catchment).
1	Baseline	/	In <u>Hestadbäcken</u> , 200 m <sup>2</sup> (0.003% of catchment) of
			floodplain & wetland. In <u>Tullstorpspån</u> , 0.08 km <sup>2</sup>
			(0.13% of catchment) of floodplain & wetland
2	20%Fert (Fert)	1	20 % reduction in mineral fertiliser application
3	Stream mitigation	3	In <u>Hestadbäcken</u> , floodplains & wetlands increase in
	(SM)		size to ca. $61,100 \text{ m}^2$ (0.80% of the catchment) with a
			barrier of 30 cm between stream. In Tullstorpån,
			floodplains & wetlands increase to 0.62 km <sup>2</sup> (1.0% of
			catchment) with a barrier of 30 cm between stream.
4	Cover crops	2	All spring crops and root crops get a cover crop in
	(CC)		between growing seasons.
5	SM+Fert	1&3	Combination of scenarios 2 and 3
6	Fert+CC	1 & 2	Combination of scenarios 2 and 4
7	SM+Fert+CC	1 & 2 & 3	Combination of scenarios 2, 3, and 4

**Table 2**: Overview of catchment mitigation scenarios

An overview of the mitigation scenarios can be found in Table 2. The selection of scenarios 210 was based on European Green Deal Ambitions, current agronomic practices in Sweden, and 211 212 stakeholder-supported stream mitigation measures in Hestadbäcken (Malgeryd et al., 2015) and Tullstorpsån (Svensson & Sundin, 2014). Impact pathway 1 (reducing nutrient inputs to fields) 213 measures were modelled by a 20% reduction in the amount of mineral fertiliser applied in a year. 214 215 Impact pathway 2 (reducing mobilisation of nutrients) measures were modelled by implementing cover crops after spring crops and root crops. Since Hestadbäcken catchment is already dominated 216 by autumn crops, this impact pathway was only modelled in Tullstorpsån. Impact pathway 3 217 (intercepting mobilised nutrient flows) measures were modelled by increasing the size of 218 floodplains & wetlands (stream mitigation scenario). Their design was modified by adding 30 cm 219 threshold barrier to decouple with the stream and retain inundation water. The baseline area of 220 floodplains & wetlands before recent stream mitigation activities was obtained from historical 221 aerial photographs. Percentage reductions of all mitigation measures were calculated against the 222 baseline situation before catchment mitigation. 223

## 3 **Results**

225 3.1 Evaluation of water quality model and climate forecasts.

226 *3.1.1 Model discharge and nutrient load calibration and validation outcomes* 

The fit of discharge and TP in Hestadbäcken can be described as 'good' to 'very good' for 227 both the calibration (2016-2020) and validation (2021-2022) periods (Table 3). For IN, the 228 Hestadbäcken fit is 'satisfactory' to 'good' in the calibration period, and 'very good' in the 229 validation period. The Pearson's correlation coefficient in both discharge and nutrient loads was 230 high (R > 0.8). There was a slight but consistent overestimation of discharge in Hestadbäcken. 231 Modelled TP loads responded opposite to discharge with a slight but consistent underestimation. 232 The bias for IN in Hestadbäcken was more variable, with alternating periods of underestimation 233 and overestimation. The modelled discharge and TP, and to a lesser extent IN, were most sensitive 234 to the wetland parameters governing its rating curve and outflow threshold (Table S2). The TP 235 loads were also sensitive to variations in sedimentation velocity. 236

In Tullstorpsån, the fit of discharge in the calibration period (2020-2022) was 'good' to 237 'very good' for both the upstream and midstream gauges, and 'very good' both during the 238 validation (2022-2023). Flow was overestimated upstream and underestimated midstream. The 239 main bias in the upstream discharge occurred during the falling limbs of high-flow events, wherein 240 modelled discharge receded slower than observed discharges. At midstream, the main bias 241 originated from missing smaller peaks in between larger events. The IN fit for downstream in 242 Tullstorpsån is 'very good' for both the calibration and validation periods, with no observed 243 systematic bias and very high correlations (R > 0.9). The fit for TP is 'very good' in the calibration 244 245 period and 'good' in the validation period, without any systematic bias. Daily flow and load comparison plots of the calibration and validation periods can be found in Figures S4-S9. 246

Table 3: Goodness-of-fit values of the calibrated model and validation period showing Nash-Sutcliffe efficiency (NSE), Kling-Gupta efficiency (KGE), Relative error (RE; %), and Pearson's correlation coefficient (R) between simulated and empirical discharges and nutrient loads. C are the calibrated years, while V are the validation years.

			Discharge			Inorganic Nitrogen			Total Phosphorus					
		Period	NSE	KGE	RE	R	NSE	KGE	RE	R	NSE	KGE	RE	R
Hestad-	C	2016-2020	0.75	0.81	14.9	0.87	0.61	0.69	-23.4	0.80	0.86	0.69	-25.9	0.93
bäcken	V	2021-2022	0.64	0.78	-10.5	0.81	0.82	0.75	-22.8	0.91	0.80	0.72	-3.0	0.95
Tullstorpsån:	C	2020-2022	0.75	0.75	22.2	0.88	/				/			
upstream	V	2022-2023	0.93	0.89	10.0	0.97	/				/			
Tullstorpsån:	C	2020-2022	0.79	0.72	-24.3	0.91	/		/					
midstream	V	2022-2023	0.93	0.87	-5.4	0.98	/				/			
Tullstorpsån:	C	2016-2022			/		0.80	0.80	-16.4	0.90	0.69	0.84	-2.0	0.85
downstream	V	2014-2015			/		0.86	0.90	0.01	0.94	0.60	0.66	-5.7	0.78

251 *3.1.2 Comparison of nutrient load forecasts under different climate models* 

In the Hestadbäcken catchment, the different downscaled climate models yielded relatively 252 similar predications in average IN loads under original conditions. The only significant or near-253 significant differences between model load quantifications were found under RCP8.5 (Table S3). 254 255 Significant differences in TP load estimations were found in 5 of the 27 model comparisons, and near-significant differences in one (Table S4). The most notable difference are the overall higher 256 estimated TP loads and their variability for the KNMI model. In the Tullstorpsån catchment, there 257 258 were slightly more divergences between model IN load predictions (Table S5), which were caused by the higher load predictions of the MPI model. Significant differences in TP load estimations 259

were found in 3 of the 27 model comparisons, and near-significant differences in another 3 (Table

- S6), wherein mostly KNMI outcomes were higher. A comparison of the empirical and predicted
- 262 precipitation and temperatures between different downscaled models is given in Text S2.
- 263 3.2 Modelled nutrient loads under different climatic trajectories

## 264 *3.2.1 Trends in projected rainfall and temperature*

Average annual rainfall and high rainfall days were predicted to increase significantly in both Hestadbäcken and Tullstorpsån under RCP8.5 (Figure S2). Under RCP2.6, the climate ensemble predicted a distinct increase in high rainfall days in 2050-2065, but no significant increase in average rainfall for both catchments. Under RCP4.5, the amount of average yearly rainfall and high rainfall days increased but not significantly. The predicted changes in temperature were more uniform between both sites (Figure S3). Under RCP4.5 and RCP8.5, significant increases of respectively 0.7 °C and 1.5 °C were predicted per period.

## 272 *3.2.2 Trends in forecasted Inorganic Nitrogen loads*

For Hestadbäcken, the ensemble model forecasted a 5% and 9% increase in IN loads under RCP2.6 for the 2050-2065 and 2085-2100 periods respectively (Figure 2). Decreases of respectively 8% and 26% in IN loads were found under RCP4.5 and decreases of respectively 23% and 50% under RCP8.5. The only trend divergence between the different climate models was observed for the ICHEC model under RCP2.6, where the loads slightly decreased in 2050-2065, and for the KNMI model under RCP4.5, where the loads remained stable in 2085-2100.

In Tullstorpsån, the ensemble model predicts no significant change under RCP2.6. Under 279 RCP4.5, the ensemble model predicts a decrease of 11% and 16% in the 2050-2065 and 2085-280 2100 respectively, while under RCP8.5 a respective decrease of 4% and 21% is forecasted. 281 However, there are some trend divergences between the models, wherein ICHEC predicts a 282 decrease in 2050-2065 under RCP2.6. Under RCP4.5, clear decreases are visible for both periods 283 284 in the ICHEC and MPI models, however, for the KNMI model, an increase in the IN load is predicted in the 2085-2100 period. Under RCP8.5, the ICHEC model predicts a strong decrease 285 followed by a slight increase, the MPI model predicts a slight increase followed by a strong 286 decrease, and the KNMI model predicts a stability followed by a strong decrease. The percentages 287 of IN load change of the ensemble and individual climate models can be found in Table S7 for 288 Hestadbäcken and Table S9 for Tullstorpsån. 289

# 290 3.2.3 Trends in forecasted Total Phosphorus loads

In Hestadbäcken, TP loads were forecasted to increase with 13% under RCP2.6 in the 291 2050-2065 period, and with 14% in 2085-2100 period, compared to the 2022-2035 period (Figure 292 2). Under RCP4.5 and RCP8.5, the ensemble climate model did not forecast any significant 293 changes in the mid nor distant future. Under RCP2.6, all models show a similar increasing trend 294 295 (5%-24%). Under RCP4.5, the KNMI showed a distinct increase in 2085-2100, while the MPI showed a distinct decrease in 2085-2100, and the ICHEC remained stable. Under RCP8.5, the 296 ICHEC and MPI models showed an increase, however with different periods of increase, while 297 298 the KNMI showed a decreasing trend.



#### (A) Hestadbäcken



**Figure 2**: Changes in nutrient loads for the (A) Hestadbäcken and (B) Tullstorpsån catchments under different climate models, the ensemble, and RCPs. Results are for the original catchment situation (see Table 2). Point and lines represent the mean yearly loads for the period, while the ribbons represent one standard deviation.

In Tullstorpsån, the ensemble model predicted no significant changes in TP loads under 304 305 RCP2.6. Under RCP4.5, the ensemble approach predicted an insignificant 3% decrease in the mid future followed by a distinct 11% increase in the distant future. This was mostly forced by the 306 KNMI model, which predicted a 9.8% increased followed by a very strong 57% increase, while 307 the other two models predicted slightly decreasing TP loads. Under RCP8.5, the ensemble model 308 predicted no significant changes in TP loads between the near and distant future. Overall, the 309 ensemble approach in Tullstorpsån did not forecast any distinct changes in TP loads over time 310 under the different RCPs. However, there was a strong divergence between the outcomes using 311 different climate models, where the KNMI model predicted significant increases under RCP4.5 312 and RCP8.5. The ICHEC model predicted increased loads under RCP2.6 and decreased loads 313 under RCP4.5 and RCP8.5. The MPI model predicted an increase followed by decrease under 314 RCP2.6, decreasing loads under RCP4.5, and increasing loads under RCP8.5. The percentage of 315 TP load change can be found in Table S8 for Hestadbäcken and Table S10 for Tullstorpsån. 316

317 3.3 Backcasted nutrient loads under catchment mitigation scenarios and climate change

## 318 *3.3.1 Hestadbäcken*

319 The stream mitigation already in place (0.03% of catchment floodplains and wetlands) only had a minor effect on nutrient load reductions in the catchment compared to the baseline scenario 320 (Figure 3). The average changes in loads range between -50.7 to 8.3% for IN and between -7.5 to 321 8.3% for TP across the RCPs and periods. Moreover, small increases in area for stream mitigation 322 will not have a strong effect on load reductions (Figure S10). A sole reduction of 20% in mineral 323 fertiliser input had strong effects on IN loads, with significant IN load reductions ranging between 324 27.0% and 68.9%. However, a reduction in mineral P inputs had no effect on the predicted TP 325 loads. A 20% reduction in mineral P fertiliser yielded load changes ranging between a 7.6% 326 reduction and a 10.8% increase, which is not significantly different from loads with current P 327 fertilisation rates. The stream mitigation scenario (floodplains & wetlands increase from 0.03% to 328 0.8%) was predicted to decrease IN loads between 39.0% and 74.4%. The stream mitigation 329 scenario was found to be the only effective measure for reducing TP loads and yielded reductions 330 ranging between 41.5% and 51.7%. Combining stream mitigation with a 20% reduction in 331 fertilisation led to the highest nutrient IN load reductions, ranging between 56.6% and 83.2%. 332 Catchment mitigation measures were also shown to decrease high nutrient load years and led to a 333 lower range in nutrient yields (Figure 3: Effects of catchment mitigation scenarios on IN and TP 334 loads under different RCPs. The boxplots depict the outcomes from the ensemble climate model, 335 where median values are shown by the central line, interquartile range by boxes, the range by 336 whiskers, and the outliers by points. The black and dotted line on the plots show the respective 337 median value and 50% reduction of the baseline scenario (no stream mitigation) in the 2022-2035 338 period.). 339

## 340 3.3.2 Tullstorpsån

The stream mitigation already in place (0.82% of catchment are floodplains and wetlands) was shown to have a significant impact of nutrient loads compared to the baseline situation before stream mitigation (0.13% of the catchment), with load reductions ranging between 6.0% and 28.1% for IN, and between 8.0% and 19.9% for TP (Figure 3). A sole reduction of 20% mineral fertiliser reduced loads ranging between 33.5% and 56.1%. The 20% reduction in mineral P fertiliser only yielded an additional 1% in TP load reduction. The stream mitigation scenario (increasing the size of floodplains & wetlands from 0.8% to 1.0%) resulted in IN load reductions

ranging between 28.9% and 50.9%, and TP load reductions between 30.4% and 41.4%. The 348 349 inclusion of cover crops reduced IN loads with 11.5% to 39.8%, but had almost no effect on TP loads. Combining a 20% decrease of mineral fertiliser and cover crops resulted in decreases of IN 350 loads ranging between 36.2% and 59.0%, and 10.4% to 21.4% decreases in TP loads. Combining 351 a 20% decrease of mineral fertiliser and stream mitigation yielded IN load decreases ranging 352 between 49.5% and 66.8%, and TP load decreases ranging between 32.0% and 42.4%. A 353 combination of a 20% fertiliser decrease, stream mitigation, and cover crops, yielded IN load 354 reductions between 51.6% and 68.9%, and TP load reductions between 32.2% and 42.5%. 355

#### 356 4 Discussion

4.1 Impact of future climate on nutrient loads in agricultural catchments

The response of nutrient loads to climate change differed between inorganic nitrogen and 358 total phosphorus and between both catchments. In Hestadbäcken, the predicated trends in IN loads 359 showed a strong similarity between the different climate models, while in Tullstorpån the 360 forecasted trends were slightly more divergent. In most cases, the differences in predicted IN loads 361 between the different climate models were smaller than between RCPs and periods, confirming 362 that our approach is robust for evaluating the impacts of future climate change on IN loads. 363 Inorganic nitrogen loads were forecasted to slightly increase under RCP2.6, decrease under 364 RCP4.5 from the distant future, and decrease under RCP8.5 from the mid future (Figure 2). These 365 diverging trends indicated dominance of increased soil N mineralisation and precipitation driven 366 IN mobilisation under RCP2.6, versus increased temperature driven denitrification and 367 evapotranspiration under RCP4.5 and RCP8.5. This indicates that future IN loads depend on the 368 interplay between the effects of increased temperature, denitrification and evapotranspiration 369 370 leading to reduced loads versus the effects of increased precipitation and IN mobilisation leading to increased loads. This pattern is supported by empirical findings of increased denitrification 371 following increasing stream temperatures in the Po river, Italy (Gervasio et al., 2022), and 372 highlights the complex feedbacks between climate change and water quality (Whitehead et al., 373 2009). Although increased temperatures in these two catchments are capable of increasing 374 denitrification, future denitrification rates will also be controlled by hydrological responses to a 375 changing rainfall distribution (Peterson et al., 2001). 376

No distinct trends in TP loads could be observed under a changing climate. The variability 377 in TP loads between the years is predicted to increase in the future (Figure 3) due to large projected 378 differences in yearly rainfall and extremes. For Hestadbäcken under RCP2.6, the model forecasted 379 that increased rainfall (intensity) will on average lead to slightly higher TP loads, most likely 380 driven by increased runoff and discharge peaks. Vice versa, under RCP4.5 and RCP8.5, the 381 projected increases in temperature and associated evapotranspiration and reduced mobilisation 382 offsetted this effect. However, the divergences in the predicted TP trends between the different 383 climate models highlighted that the uncertainty of these TP load outcomes remained large in both 384 catchments. 385

Overall, the complex responses of IN and TP loads to different RCPs and forecasted periods demonstrated the challenges of predicting nutrient losses and eutrophication in agricultural catchments under a changing climate. The predicted increase in nutrient loads in the near future will require immediate mitigation actions to be undertaken to achieve EU green deal targets. These mitigation measures will need to focus on buffering high flow events with large mobilisation of nutrients. Water quality management in the medium to distant future will be required to adapt in response to the periodic climatic conditions.

## 393 4.2 Mitigating future nutrient loads in light of EU Green Deal targets

In both catchments, target 50% reductions in both nutrients could only be achieved by 394 including the stream mitigation scenario. Since proposed activities under the Water Framework 395 Directive and EU Green deal focus on land measures, our results highlight the importance of 396 tackling secondary pollution sources using stream mitigation (M. Bieroza et al., 2021). Yearly TP 397 loads in the study catchments were largely controlled by mobilisation of phosphorus in the stream, 398 irrespective of reduced phosphorus fertilisation or inclusion of cover crops. Mobilisation of 399 phosphorus from eroding stream banks (Fox et al., 2016) or by resuspension of bed sediments 400 (Ballantine et al., 2009) have been shown to be important drivers of TP loads. Field observations 401 also confirm that Hestadbäcken and the upstream reaches of Tullstorpsån are characterised by large 402 stores of fine sediments. The lack of TP load response to a reduction in phosphorus fertilisation 403 corresponded with results by Capell et al. (2021) in the Baltic sea region, but contrasted with the 404 outcomes from Ockenden et al. (2017) in England. The strong binding of phosphate to fine-405 textured soils and sediments in the study sites might form an explanation (Sandström et al., 2021). 406 Moreover, legacy phosphorus pools, both in organic form and attached to soil particles, in Swedish 407 topsoils are large compared to annual fertiliser rates (Capell et al., 2021). In this study, the legacy 408 phosphorus thus overrode the impacts of reduced phosphorus fertilisation. However, reduction of 409 phosphorus fertiliser might still be valuable long-term measure in combination with circular 410 management of available legacy stores (Haygarth et al., 2014; Withers et al., 2014). We therefore 411 suggest that water quality models be supplemented with new routines to simulate draw down of 412 legacy nutrient levels through agronomic practices. 413

Stream mitigation was also effective in reducing IN loads in both study catchments. In 414 Hestadbäcken, the stream mitigation scenario was sufficient to reduce both IN and TP loads by 415 50% in most combinations of RCPs and periods compared to the baseline. We found that for target 416 50% IN load reductions in Hestadbäcken, stream mitigation would need to be complemented with 417 a 20% reduction in mineral fertilisation in the 2022-2035 period. However, since climate change 418 419 will have a strong impact on the required mitigation measures for IN, the reduction in mineral nitrogen fertiliser could be stopped from the mid future under RCP4.5 and RCP8.5. In 420 Tullstorpsån, the tested stream mitigation scenario did not achieve the targeted 50% reduction of 421 TP, but combining stream mitigation with a 20% reduction in IN fertiliser led to the targeted 50% 422 423 reduction of IN loads. Under RCP8.5 in 2050-2065 and 2085-2100, stream mitigation could also be combined with the use of cover crops to reduce IN loads with approximately 50%. 424

425 The different strength of nutrient loads reductions to the tested mitigation scenarios between the two study catchments can be explained by multiple environmental factors. 426 Hestadbäcken is smaller, drier, and has a higher hydrological connectivity compared to 427 Tullstorpsån, thus will respond stronger to changes in driving climatic factors. This also explains 428 429 the stronger effect of stream mitigation in Hestadbäcken, which acts as a buffer for discharge and nutrient flows. The coarser texture of agricultural soils in Tullstorpån explains the slight response 430 to the reduction in mineral phosphorus, compared to no response in Hestadbäcken. These findings 431 indicate that a uniform 50% reduction in nutrients might not be feasible and mitigation measures 432 should target the highest value for money for each catchment. In the context of these multiple 433 spatial factors, further research is needed to determine to what degree these findings and 434 recommendations can be extrapolated to larger geographical regions. 435



436

Figure 3: Effects of catchment mitigation scenarios on IN and TP loads under different RCPs. The boxplots depict the outcomes from the ensemble climate model, where median values are shown

by the central line, interquartile range by boxes, the range by whiskers, and the outliers by points. The black and dotted line on the plots show the respective median value and 50% reduction of the

441 baseline scenario (no stream mitigation) in the 2022-2035 period.

442 4.3 Trade-offs and synergies in mitigating nutrient losses

We argue that stream mitigation is critical for reaching water quality targets in the study 443 sites because it is the only measure effective for reducing both TP and IN loads. However, our 444 findings also highlight the importance of the size of stream mitigation zones for achieving 445 significant reductions in nutrient loads (Fig. S10). In both study sites, nearly 1% of the catchment 446 should thus be set aside for floodplains and wetlands to achieve set targets, which corresponds 447 with proposed guidelines for Sweden (Arheimer & Pers, 2017). In Hestadbäcken, the stream 448 mitigation scenario would amount to an area of roughly 61,500 m<sup>2</sup>, which would require a stream 449 length of ca. 3,100 m with an average width of ca. 20 m. However, this effect is catchment specific 450 as in Tullstorpån the stream mitigation scenario (additional area of  $110,000 \text{ m}^2$ , which corresponds 451 to ca. 6,100 m of unmitigated stream length and an average width of ca. 18 m) would still not allow 452 to reach the 50% reduction in TP loads. Modelled stream mitigation scenarios are implemented 453 downstream and are thus likely to reduce productive arable land. As implementation of any 454 mitigation measures is voluntary among landowners, stream remediation could be avoided by 455 landowners who would prefer more soil-based measures, such as cover crops, that have less impact 456 on their crop production. A clear estimation of cost-effectiveness of different measures is thus 457 needed. This should be also supported by financial incentives for landowners who would be 458 required to lose productive land in order to implement more effective stream mitigation (Bol et al., 459 2018). When estimating cost-effectiveness other ecosystem services enabled by stream mitigation 460 such as flood control, biodiversity improvement, and recreation should be taken into account 461 (Hambäck et al., 2023). Stream mitigation can achieve a higher cost-effectiveness compared to 462 measures strictly focusing on one ecosystem service or water quality problem. Moreover, more 463 stringent guidelines and requirements on minimum size and design of stream mitigation are needed 464 to achieve maximum impact and avoid too narrow stream mitigation zones (Arheimer & Pers, 465 2017; Noe et al., 2013). 466

An important consideration is the combined effect of multiple measures. In our study, the 467 468 IN load reductions when combining multiple mitigation measures were lower than the summed reductions of the individual measures, indicating that the impact of these measures is not 469 synergistic (M. Bieroza et al., 2019). This is likely because the efficacy of stream mitigation for 470 nutrient retention and denitrification is also partly determined by the incoming nutrient loads 471 (Hallberg et al., 2022; Noe et al., 2013), which will be lower with reduced fertilisation or cover 472 crops. While catchment mitigation measures were found to be effective in reducing total nutrient 473 474 loads, they were also effective in buffering the nutrient loads during high rainfall years, evidenced by the lower overall variability and range in nutrient loads. This is particularly pronounced for 475 stream mitigation in years with high rainfall (Figure 3), which is important since years with high 476 nutrient loads can destabilise aquatic ecosystems even if average nutrient loads do not increase (A. 477 Yang et al., 2022). 478

479 4.4 Uncertainty in modelled nutrient loads

Overall, the models performed well to predict flow discharge and nutrient loads, both on event-scale and yearly scale. The main source of error in discharge is likely because both catchments are dominated by crop cultivation with associated soil management and tile drainage. The drainage efficiency and depth of tile drains was assumed similar in all SLCs, while it is highly variable in reality. Small-scale rainfall events are also not always well represented in the meteorological inputs to the model and this might be a likely explanation for the absence of some discharge events in the model. The overestimation of discharge in the upstream part of Tullstorpsån and underestimation in the midstream part also indicated challenges in representing hydrologicalconnectivity on different spatial scales.

Stochastic contribution of point sources, such as rural sewage or runoff from livestock 489 stables, are a likely overall source of error in the nutrient loads. The most likely explanation for 490 the underestimation of simulated TP in Hestadbäcken are the large amounts of fine sediment in 491 the stream, which could act as a store of easily mobilised phosphorus. Moreover, the effects of 492 high intensity rainfall events are not fully captured in the daily timestep of the model. The observed 493 underestimation of modelled IN, especially in Hestadbäcken, might be caused by legacy IN and 494 changing importance of subsurface delivery to the river discharge. Subsurface water discharged 495 from tile drains and groundwater contributes more proportionally to the base flows and the falling 496 limbs of storm events, thereby disproportionally increasing the IN loads (M. Z. Bieroza et al., 497 2018). Finally, the model also did not always pick up the 'first flush effects' (September 2018 in 498 Hestadbäcken), wherein a dry summer led to lower nutrient uptake by crops, as well as exposure 499 of dried out stream bed sediments, resulting in disproportionally high nutrient loads during the first 500 high flow event (M. Bieroza et al., 2019). The larger catchment area and higher average 501 precipitation throughout the year in Tullstorpsån make it less vulnerable to these effects. 502

This latter issue exposes a limitation of HYPE in flashy headwater catchments, wherein 503 mineralisation of bed sediments and crop nutrient uptake are not influenced by rainfall and 504 hydrology. It is expected that potential crop growth in Sweden will increase under the predicted 505 higher temperatures due to a longer growing season (Wiréhn, 2018). However, since crop nutrient 506 uptake can be potentially limited by low water availability during dry summers (Grusson et al., 507 2021), this could result in higher nutrient loads in winter. Another limitation of HYPE is in its 508 representation of floodplains & wetlands, which are modelled as nutrient buffers between the soil 509 and river. In reality, floodplains & wetlands are characterised by dynamic water tables and redox 510 conditions, which influences denitrification and nutrient mobility. Moreover, stream mitigation 511 influences stream hydraulics and thus sediment and nutrient transport dynamics (Noe et al., 2013). 512 Moreover, stream and wetland parameters were calibrated for their current size, leading to 513 potential unrealistic modulations in the stream mitigation scenarios. However, the sensitivity 514 analysis revealed that only large changes in the rating curve and outflow threshold parameters 515 resulted into significant NSE variance for discharge and TP loads. Since the outflow threshold was 516 not calibrated but based on field observations or scenarios, and the rating curve parameters were 517 strongly constrained, we argue that the uncertainty around the modelled impacts of stream 518 mitigation is miminal. The used ensemble climate forecast approach provided an uncertainty range 519 of the forecasted nutrient loads, which remained particularly large for TP loads driven by the higher 520 predicted rainfall in the KNMI model. Uncertainties are expected to decrease with improved 521 climate forecasting under CMIP6 (Jacob et al., 2020; Krysanova et al., 2018). In the context of 522 these complex interrelations between climate dynamics, vegetation growth, and nutrient dynamics, 523 a promising next step in modelling future water quality would be to couple hydrological models 524 525 such as HYPE with other disciplinary models (e.g. hydraulic, dynamic crop growth, nutrient legacies, and land use), into a systems dynamic model (Duran-Encalada et al., 2017). 526

## 527 **5 Conclusions**

This study forecasted nutrient load exports up to year 2100 using three climate trajectories. IN loads were predicted to decrease under RCP4.5 and RCP8.5 due to increased denitrification and evapotranspiration. Under RCP2.6, IN loads were forecasted to increase in Hestadbäcken, while remaining stable in Tullstorpsån. The response of TP loads to climate change was found to be highly variable and a significant increase only occurred under RCP2.6 in Hestadbäcken. These
 findings highlighted the divergent responses of IN and TP dynamics to climate change.

Moreover, this work successfully demonstrated a methodology for backcasting catchment 534 mitigation scenarios to achieve the European Green Deal ambition of 50% reduction in nutrient 535 exports from agricultural catchments under a changing climate. A reduction in mineral fertilisation 536 is highly effective for reducing IN loads, but has almost no effects on TP loads. Likewise, cover 537 crops showed a promising effect for reducing IN loads, but have almost no effect on TP loads. 538 Increasing the size and design of floodplains & wetlands reduces total export of both IN and TP. 539 These outcomes are likely due to the dominance of stream processes for mobilising TP and large 540 legacy phosphorus stores in soils and sediments. Since TP load reductions only respond to the 541 floodplains & wetlands scenarios, we argue that in these two cases they are critical for catchment 542 mitigation plans. Overall, the diverging outcomes highlight that the optimal mitigation scenarios 543 are dependent on land use, soil type, nutrient form, and the spatial and temporal effects of a 544 changing climate. This study has demonstrated the potential of catchment water quality modelling 545 as a first step in decision support to find the most effective ways to mitigate nutrient loads in 546 agricultural catchments. 547

## 548 Acknowledgments

This research was jointly funded by Formas, SLF, Stiftelsen Oscar och Lili Lamms Minne, 549 and Baltic Waters. We thank Katarina Kyllmar, Maria Blomberg and Helena Linefur from the 550 Swedish University of Agricultural Sciences for curating and sharing the stream monitoring data 551 from Hestadbäcken. We would also like to thank Anuschka Heeb from Lovang Lantbrukskonsult 552 AB for sharing agronomic practices in Hestadbäcken and Christoffer Bonthron from the 553 Tullstorpsån Project (www.tullstorpsan.se) for sharing the stream chemistry data from the 554 downstream site. Finally, we would like to thank Charlotta Pers, René Capell, Kristina Isberg, 555 Conrad Brendel, and Alena Bartosova from SMHI for their support with HYPE and HYPE tools. 556

## 557 **Open Research**

All processed geospatial data, water quality and discharge data used for calibration, climate data, model set-up, and model outcomes can be previewed by reviewers on the Wynants et al. (2023) dataset. The HYPE model and HYPE tools can be accessed open access via respectively https://sourceforge.net/projects/hype/files/ and https://github.com/rcapell/HYPEtools.

## 562 **References**

- Andersen, J. H., Carstensen, J., Conley, D. J., Dromph, K., Fleming-Lehtinen, V., Gustafsson, B. G., et al. (2017).
   Long-term temporal and spatial trends in eutrophication status of the Baltic Sea. *Biological Reviews*, 92(1),
   135-149.
- Arheimer, B., & Pers, B. C. (2017). Lessons learned? Effects of nutrient reductions from constructing wetlands in
   1996–2006 across Sweden. *Ecological Engineering*, 103, 404-414.
- Arheimer, B., & Wittgren, H. B. (2002). Modelling nitrogen removal in potential wetlands at the catchment scale.
   *Ecological engineering*, 19(1), 63-80.
- Ballantine, D. J., Walling, D. E., Collins, A. L., & Leeks, G. J. L. (2009). The content and storage of phosphorus in
   fine-grained channel bed sediment in contrasting lowland agricultural catchments in the UK. *Geoderma*,
   151(3), 141-149.

- Bartosova, A., Arheimer, B., De Lavenne, A., Capell, R., & Strömqvist, J. (2021). Large-Scale Hydrological and
   Sediment Modeling in Nested Domains under Current and Changing Climate. *Journal of Hydrologic Engineering*, 26(5), 05021009.
- Bartosova, A., Capell, R., Olesen, J. E., Jabloun, M., Refsgaard, J. C., Donnelly, C., et al. (2019). Future
  socioeconomic conditions may have a larger impact than climate change on nutrient loads to the Baltic Sea. *Ambio*, 48, 1325-1336.
- Basu, N. B., Dony, J., Van Meter, K., Johnston, S. J., & Layton, A. T. (2023). A random forest in the Great Lakes:
  Stream nutrient concentrations across the transboundary Great Lakes Basin. *Earth's Future*, *11*(4),
  e2021EF002571.
- Basu, N. B., Van Meter, K. J., Byrnes, D. K., Van Cappellen, P., Brouwer, R., Jacobsen, B. H., et al. (2022).
   Managing nitrogen legacies to accelerate water quality improvement. *Nature Geoscience*, 15(2), 97-105.
- Berg, P., Norin, L., & Olsson, J. (2016). Creation of a high resolution precipitation data set by merging gridded
   gauge data and radar observations for Sweden. *Journal of Hydrology*, *541*, 6-13.
- Bieroza, M., Bergström, L., Ulén, B., Djodjic, F., Tonderski, K., Heeb, A., et al. (2019). Hydrologic extremes and
   legacy sources can override efforts to mitigate nutrient and sediment losses at the catchment scale. *Journal of Environmental Quality*, 48(5), 1314-1324.
- Bieroza, M., Bol, R., & Glendell, M. (2021). What is the deal with the Green Deal: Will the new strategy help to
   improve European freshwater quality beyond the Water Framework Directive? Science of the Total
   *Environment*, 791, 148080.
- Bieroza, M. Z., Heathwaite, A. L., Bechmann, M., Kyllmar, K., & Jordan, P. (2018). The concentration-discharge
   slope as a tool for water quality management. *Science of the Total Environment*, *630*, 738-749.
- Blann, K. L., Anderson, J. L., Sands, G. R., & Vondracek, B. (2009). Effects of agricultural drainage on aquatic
   ecosystems: a review. *Critical Reviews in Environmental Science and Technology*, 39(11), 909-1001.
- Boeuf, B., & Fritsch, O. (2016). Studying the implementation of the Water Framework Directive in Europe: a metaanalysis of 89 journal articles. *Ecology and Society*, 21(2).
- Bol, R., Gruau, G., Mellander, P.-E., Dupas, R., Bechmann, M., Skarbøvik, E., et al. (2018). Challenges of reducing
   phosphorus based water eutrophication in the agricultural landscapes of Northwest Europe. *Frontiers in Marine Science*, 5, 276.
- Bouwman, L., Goldewijk, K. K., Van Der Hoek, K. W., Beusen, A. H., Van Vuuren, D. P., Willems, J., et al.
  (2013). Exploring global changes in nitrogen and phosphorus cycles in agriculture induced by livestock
  production over the 1900–2050 period. *Proceedings of the National Academy of Sciences, 110*(52), 2088220887.
- Braak, C. J. T. (2006). A Markov Chain Monte Carlo version of the genetic algorithm Differential Evolution: easy
   Bayesian computing for real parameter spaces. *Statistics and Computing*, *16*, 239-249.
- Capell, R., Bartosova, A., Tonderski, K., Arheimer, B., Pedersen, S. M., & Zilans, A. (2021). From local measures
   to regional impacts: Modelling changes in nutrient loads to the Baltic Sea. *Journal of Hydrology: Regional Studies, 36*, 100867.
- Castellano, M. J., Archontoulis, S. V., Helmers, M. J., Poffenbarger, H. J., & Six, J. (2019). Sustainable
   intensification of agricultural drainage. *Nature sustainability*, 2(10), 914-921.
- Collins, M., Knutti, R., Arblaster, J., Dufresne, J.-L., Fichefet, T., Friedlingstein, P., et al. (2013). Long-term climate
  change: projections, commitments and irreversibility. In *Climate Change 2013: The Physical Science Basis. IPCC Working Group I Contribution to AR5* (pp. 1029-1136). Cambridge, UK and New York, USA:
  Cambridge University Press, .
- Destouni, G., Fischer, I., & Prieto, C. (2017). Water quality and ecosystem management: Data-driven reality check
   of effects in streams and lakes. *Water Resources Research*, 53(8), 6395-6406.
- Duran-Encalada, J. A., Paucar-Caceres, A., Bandala, E., & Wright, G. (2017). The impact of global climate change
   on water quantity and quality: A system dynamics approach to the US–Mexican transborder region.
   *European Journal of Operational Research*, 256(2), 567-581.
- 621 ESRI. (2020). ArcGIS Desktop: Release 10.8.1. Redlands, CA: Environmental Systems Research Institute.
- European Commission. (2020). Farm to fork strategy: for a fair, healthy and environmentally-friendly food system.
   Retrieved from <a href="https://agriculture.ec.europa.eu/sustainability/environmental-sustainability/low-input-farming/nutrients\_en">https://agriculture.ec.europa.eu/sustainability/environmental-sustainability/low-input-farming/nutrients\_en</a>
- Fox, G. A., Purvis, R. A., & Penn, C. J. (2016). Streambanks: A net source of sediment and phosphorus to streams
   and rivers. *Journal of Environmental Management*, 181, 602-614.
- Frei, R. J., Lawson, G. M., Norris, A. J., Cano, G., Vargas, M. C., Kujanpää, E., et al. (2021). Limited progress in nutrient pollution in the US caused by spatially persistent nutrient sources. *PLoS ONE*, 16(11), e0258952.

- Gervasio, M. P., Soana, E., Granata, T., Colombo, D., & Castaldelli, G. (2022). An unexpected negative feedback
  between climate change and eutrophication: higher temperatures increase denitrification and buffer
  nitrogen loads in the Po River (Northern Italy). *Environmental Research Letters*, 17(8), 084031.
- Grimvall, A., von Brömssen, C., & Lindström, G. (2014). Using process-based models to filter out natural
   variability in observed concentrations of nitrogen and phosphorus in river water. *Environmental monitoring and assessment*, 186, 5135-5152.
- Grusson, Y., Wesström, I., Svedberg, E., & Joel, A. (2021). Influence of climate change on water partitioning in
   agricultural watersheds: Examples from Sweden. *Agricultural Water Management*, 249, 106766.
- Gupta, H. V., Kling, H., Yilmaz, K. K., & Martinez, G. F. (2009). Decomposition of the mean squared error and
   NSE performance criteria: Implications for improving hydrological modelling. *Journal of Hydrology*,
   377(1-2), 80-91.
- Hallberg, L., Hallin, S., & Bieroza, M. (2022). Catchment controls of denitrification and nitrous oxide production
   rates in headwater remediated agricultural streams. *Science of the Total Environment*, 838, 156513.
- Hambäck, P., Dawson, L., Geranmayeh, P., Jarsjö, J., Kačergytė, I., Peacock, M., et al. (2023). Tradeoffs and
   synergies in wetland multifunctionality: A scaling issue. *Science of the Total Environment*, 862, 160746.
- Hankin, B., Strömqvist, J., Burgess, C., Pers, C., Bielby, S., Revilla-Romero, B., & Pope, L. (2019). A new national
  water quality model to evaluate the effectiveness of catchment management measures in England. *Water*,
  11(8), 1612.
- Haygarth, P. M., Jarvie, H. P., Powers, S. M., Sharpley, A. N., Elser, J. J., Shen, J., et al. (2014). Sustainable
   phosphorus management and the need for a long-term perspective: The legacy hypothesis. *Environmental Science & Technology*, 48(15), 8417-8419.
- Hazeleger, W., Severijns, C., Semmler, T., Ştefănescu, S., Yang, S., Wang, X., et al. (2010). EC-Earth: a seamless
  earth-system prediction approach in action. *Bulletin of the American Meteorological Society*, *91*(10), 13571364.
- Hogan, S. C., O'Sullivan, J. J., Bruen, M., Jarvie, H. P., Cox, E. J., Bowes, M. J., & Kelly-Quinn, M. (2023). A
   review of the nature and source of nutrient impairment in small streams: a desk based characterisation for
   targeted mitigation measures. *Hydrobiologia*, 1-19.
- Hundecha, Y., Arheimer, B., Berg, P., Capell, R., Musuuza, J., Pechlivanidis, I., & Photiadou, C. (2020). Effect of
   model calibration strategy on climate projections of hydrological indicators at a continental scale. *Climatic Change*, 163, 1287-1306.
- Jacob, D., Petersen, J., Eggert, B., Alias, A., Christensen, O. B., Bouwer, L. M., et al. (2014). EURO-CORDEX:
   new high-resolution climate change projections for European impact research. *Regional Environmental Change, 14*, 563-578.
- Jacob, D., Teichmann, C., Sobolowski, S., Katragkou, E., Anders, I., Belda, M., et al. (2020). Regional climate
   downscaling over Europe: perspectives from the EURO-CORDEX community. *Regional Environmental Change*, 20, 1-20.
- Jones, C., Hughes, J., Bellouin, N., Hardiman, S., Jones, G., Knight, J., et al. (2011). The HadGEM2-ES
   implementation of CMIP5 centennial simulations. *Geoscientific Model Development*, 4(3), 543-570.
- Jordbruksverket. (2022). Mark- och grödoinventeringen samt Nationell jordartskartering. Retrieved 2022
   <u>https://miljodata.slu.se/mvm/aker</u>
- Knoben, W. J., Freer, J. E., & Woods, R. A. (2019). Inherent benchmark or not? Comparing Nash–Sutcliffe and
   Kling–Gupta efficiency scores. *Hydrology and Earth System Sciences*, 23(10), 4323-4331.
- Kristensen, P., Whalley, C., Zal, F. N. N., & Christiansen, T. (2018). *European waters assessment of status and pressures 2018* (1977-8449). Retrieved from <u>https://www.eea.europa.eu/publications/state-of-water</u>
- Krysanova, V., Donnelly, C., Gelfan, A., Gerten, D., Arheimer, B., Hattermann, F., & Kundzewicz, Z. W. (2018).
   How the performance of hydrological models relates to credibility of projections under climate change.
   *Hydrological Sciences Journal*, 63(5), 696-720.
- Kyllmar, K., Forsberg, L. S., Andersson, S., & Mårtensson, K. (2014). Small agricultural monitoring catchments in
   Sweden representing environmental impact. *Agriculture, Ecosystems & Environment, 198*, 25-35.
- Lantmäteriet. (2019). GSD-Elevation data, Grid 2+. Retrieved from <a href="https://www.lantmateriet.se/sv/Kartor-och-geografisk-information/geodataprodukter/hojddata-grid-2">https://www.lantmateriet.se/sv/Kartor-och-geografisk-information/geodataprodukter/hojddata-grid-2</a>
- Lantmäteriet. (2021). Ortofoto Nedladdning. Retrieved from <u>https://www.lantmateriet.se/en/geodata/geodata-geodata/geodat</u>
- Lindström, G., Pers, C., Rosberg, J., Strömqvist, J., & Arheimer, B. (2010). Development and testing of the HYPE
   (Hydrological Predictions for the Environment) water quality model for different spatial scales. *Hydrology Research*, 41(3-4), 295-319.

- Lu, C., & Tian, H. (2017). Global nitrogen and phosphorus fertilizer use for agriculture production in the past half
   century: shifted hot spots and nutrient imbalance. *Earth System Science Data*, 9(1), 181-192.
- Malgeryd, J., Stjernman-Forsberg, L., Kyllmar, K., Heeb, A., Gustafsson, J., & Svensson, A. (2015). Åtgärder mot
   fosforförluster från jordbruksmark—erfarenheter från tre avrinningsområden i Västmanland, Östergötland
   och Halland. Slutrapport och delrapport 2 från projekt Greppa Fosforn, 2010–2014. *Swedish.*)
   *Jordbruksverket Rapport*, 2.
- Mellander, P.-E., Jordan, P., Bechmann, M., Fovet, O., Shore, M. M., McDonald, N. T., & Gascuel-Odoux, C.
   (2018). Integrated climate-chemical indicators of diffuse pollution from land to water. *Scientific Reports*, 8(1), 944.
- Moriasi, D. N., Gitau, M. W., Pai, N., & Daggupati, P. (2015). Hydrologic and water quality models: Performance
   measures and evaluation criteria. *Transactions of the ASABE*, 58(6), 1763-1785.
- Nash, J. E., & Sutcliffe, J. V. (1970). River flow forecasting through conceptual models part I—A discussion of principles. *Journal of Hydrology*, *10*(3), 282-290.
- Noe, G. B., Hupp, C. R., & Rybicki, N. B. (2013). Hydrogeomorphology influences soil nitrogen and phosphorus
   mineralization in floodplain wetlands. *Ecosystems*, 16(1), 75-94.
- Ockenden, M. C., Hollaway, M. J., Beven, K. J., Collins, A., Evans, R., Falloon, P., et al. (2017). Major agricultural
   changes required to mitigate phosphorus losses under climate change. *Nature Communications*, 8(1), 161.
- Peterson, B. J., Wollheim, W. M., Mulholland, P. J., Webster, J. R., Meyer, J. L., Tank, J. L., et al. (2001). Control of nitrogen export from watersheds by headwater streams. *Science*, 292(5514), 86-90.
- Piikki, K., & Söderström, M. (2019). Digital soil mapping of arable land in Sweden–Validation of performance at multiple scales. *Geoderma*, 352, 342-350.
- Popke, D., Stevens, B., & Voigt, A. (2013). Climate and climate change in a radiative-convective equilibrium
   version of ECHAM6. *Journal of Advances in Modeling Earth Systems*, 5(1), 1-14.
- R Core Team. (2022). R: A language and environment for statistical computing. Vienna, Austria. Retrieved from <a href="http://www.R-project.org">http://www.R-project.org</a>
- Roley, S. S., Tank, J. L., Tyndall, J. C., & Witter, J. D. (2016). How cost-effective are cover crops, wetlands, and
   two-stage ditches for nitrogen removal in the Mississippi River Basin? *Water Resources and Economics*,
   15, 43-56.
- Sandström, S., Futter, M. N., O'Connell, D. W., Lannergård, E. E., Rakovic, J., Kyllmar, K., et al. (2021).
   Variability in fluvial suspended and streambed sediment phosphorus fractions among small agricultural streams. *Journal of Environmental Quality*, 50(3), 612-626.
- Santos, L., Andersson, J. C., & Arheimer, B. (2022). Evaluation of parameter sensitivity of a rainfall-runoff model
   over a global catchment set. *Hydrological Sciences Journal*, 67(3), 342-357.
- Schultz, B., Zimmer, D., & Vlotman, W. F. (2007). Drainage under increasing and changing requirements.
   *Irrigation and Drainage: The journal of the International Commission on Irrigation and Drainage*, 56(S1),
   S3-S22.
- Strandberg, G., Bärring, L., Hansson, U., Jansson, C., Jones, C., Kjellström, E., et al. (2015). CORDEX scenarios
   *for Europe from the Rossby Centre regional climate model RCA4*. Retrieved from
   <u>https://www.smhi.se/en/publications/cordex-scenarios-for-europe-from-the-rossby-centre-regional-climate-</u>
   <u>model-rca4-1.90274</u>
- Strömqvist, J., Arheimer, B., Dahné, J., Donnelly, C., & Lindström, G. (2012). Water and nutrient predictions in ungauged basins: set-up and evaluation of a model at the national scale. *Hydrological Sciences Journal*, 57(2), 229-247.
- Svensson, E., & Sundin, S. (2014). The Tullstorp Stream Project: Success factors, challenges and recommendations for improvement of agri-environmental projects. Retrieved from
- 730 <u>https://tullstorpsan.se/rapporter/CasestudyforBalticCompact141223.pdf</u>
   731 Sveriges Geologiska Undersökning, (2014). JORDARTER 1:25 000-1:100 0000
- 731 Sveriges Geologiska Undersokning. (2014). JORDARTER 1:25 000-1:100 0000
- Swedish Meteorological and Hydrological Institute. (2023a). Griddade nederbörd- och temperaturdata PTHBV.
   Retrieved 2023 <u>https://www.smhi.se/data/ladda-ner-data/griddade-nederbord-och-temperaturdata-pthbv</u>
- Swedish Meteorological and Hydrological Institute. (2023b). Meteorologiska observationer. Retrieved 2023
   <u>https://www.smhi.se/data/meteorologi/ladda-ner-meteorologiska-</u>
   <u>observationer#param=airtemperatureInstant,stations=core,stationid=86340</u>
- Tonderski, K. S., Arheimer, B., & Pers, C. B. (2005). Modeling the impact of potential wetlands on phosphorus
   retention in a Swedish catchment. *AMBIO: A Journal of the Human Environment*, 34(7), 544-551.

- Ulén, B., Bechmann, M., Fölster, J., Jarvie, H., & Tunney, H. (2007). Agriculture as a phosphorus source for
   eutrophication in the north-west European countries, Norway, Sweden, United Kingdom and Ireland: a
   review. *Soil use and Management*, 23, 5-15.
- Uusi-Kämppä, J., Braskerud, B., Jansson, H., Syversen, N., & Uusitalo, R. (2000). Buffer zones and constructed
   wetlands as filters for agricultural phosphorus. *Journal of Environmental Quality*, 29(1), 151-158.
- van Meijgaard, E., Van Ulft, L., Van de Berg, W., Bosveld, F., Van den Hurk, B., Lenderink, G., & Siebesma, A.
   (2008). *The KNMI regional atmospheric climate model RACMO, version 2.1*. Retrieved from De Bilt, The
   Netherlands: <u>https://cdn.knmi.nl/knmi/pdf/bibliotheek/knmipubTR/TR302.pdf</u>
- Walton, C. R., Zak, D., Audet, J., Petersen, R. J., Lange, J., Oehmke, C., et al. (2020). Wetland buffer zones for
   nitrogen and phosphorus retention: Impacts of soil type, hydrology and vegetation. *Science of the Total Environment*, 727, 138709.
- Whitehead, P. G., Wilby, R. L., Battarbee, R. W., Kernan, M., & Wade, A. J. (2009). A review of the potential impacts of climate change on surface water quality. *Hydrological Sciences Journal*, 54(1), 101-123.
- Wiering, M., Liefferink, D., Boezeman, D., Kaufmann, M., Crabbé, A., & Kurstjens, N. (2020). The wicked
   problem the water framework directive cannot solve. The governance approach in dealing with pollution of
   nutrients in surface water in The Netherlands, Flanders, Lower Saxony, Denmark and Ireland. *Water*,
   12(5), 1240.
- Wiréhn, L. (2018). Nordic agriculture under climate change: A systematic review of challenges, opportunities and adaptation strategies for crop production. *Land use policy*, *77*, 63-74.
- Withers, P. J., Sylvester-Bradley, R., Jones, D. L., Healey, J. R., & Talboys, P. J. (2014). Feed the crop not the soil:
   rethinking phosphorus management in the food chain. *Environmental Science & Technology*, 48(12), 6523-6530.
- Wynants, M., Hallberg, L., Livsey, J., & Bieroza, M. (2023). Dataset for paper "How to achieve a 50% reduction in nutrient loads from agricultural catchments under different climate trajectories?". Retrieved from: https://snd.gu.se/en/catalogue/study/preview/791724f4-a6ab-4ee7-831e-23e636d2f4d8#dataset
- Wynants, M., Solomon, H., Ndakidemi, P., & Blake, W. H. (2018). Pinpointing areas of increased soil erosion risk
   following land cover change in the Lake Manyara catchment, Tanzania. *International Journal of Applied Earth Observation and Geoinformation*, 71, 1-8.
- Yang, A., Wang, H., Zhang, T., & Yuan, S. (2022). Stochastic switches of eutrophication and oligotrophication:
   Modeling extreme weather via non-Gaussian Lévy noise. *Chaos: An Interdisciplinary Journal of Nonlinear Science*, *32*(4), 043116.
- Yang, W., Andréasson, J., Phil Graham, L., Olsson, J., Rosberg, J., & Wetterhall, F. (2010). Distribution-based
   scaling to improve usability of regional climate model projections for hydrological climate change impacts
   studies. *Hydrology Research*, 41(3-4), 211-229.
- Zia, A., Schroth, A. W., Hecht, J. S., Isles, P., Clemins, P. J., Turnbull, S., et al. (2022). Climate Change-Legacy
   Phosphorus Synergy Hinders Lake Response to Aggressive Water Policy Targets. *Earth's Future*, 10(5),
   e2021EF002234.
- 776